Questions of the Reviewers:

1: There are often thickened and disorganized blood vessels around sarcomas. Is the vascular malformation in this case caused by stromal tumors? After all, the patient does not have symptoms of gastrointestinal bleeding?

—› In the pathological examination we could see that the vascular malformation could be explained by the striomal tumor. Since it is a rare but known syndrome we conclude that the vascular malformation is caused by the stormal tumor. The patient did not have any symptoms of G.I. bleeding.

2: Did the author used the methods described in this article to handle cases of vascular malformations which caused the symptoms of gastrointestinal bleeding before?

—› No we did not use this techninque before. We used this technique in transplant surgery to see the perfusion of the allografts. Since we could monitor this with the hyperspectral camera we used this technique in oncology surgery.